Author's response to reviews

Title: Effects of two Lactobacillus strains on lipid metabolism and intestinal microflora in rats fed a high-cholesterol diet

Authors:

Ning Xie (laziowin@126.com)
Yi Cui (cuiyi1000@126.com)
Ya-Ni Yin (niniyaya123@sina.com)
Xin Zhao (109860810@qq.com)
Jun-Wen Yang (695399512@qq.com)
Zheng-Gen Wang (wzg987@yeah.net)
Nian Fu (fulian365@sina.com)
Yong Tang (1037411437@qq.com)
Xue-Hong Wang (wxh05271@sina.com)
Xiao-Wei Liu (liuxiaoweiabc@126.com)
Chun-Lian Wang (wcl0747@163.com)
Fang-Gen Lu (lufanggenyao@163.com)

Version: 3 Date: 24 June 2011

Author's response to reviews:

1. Each affiliation had been written in full according to the format (since the streets are not compulsory, we omitted them).
2. Use initials instead of full name in the authors' contributions section.
3. All references had been checked.
4. Line numbers and underlinings were removed.
5. Tables were checked, you can make more changes if necessary. Thanks.
6. The figures were cropped as closely as possible.